Author's response to reviews

Title: Endocarditis caused by methicillin-susceptible Staphylococcus aureus with reduced susceptibility to vancomycin: a case report.

Authors:

Beatriz E Perazzi (hugodandrea@ciudad.com.ar)
Natalia Bello (nataliasbello@yahoo.com.ar)
Marta Mollerach (martamollerach@gmail.com)
Carlos A Vay (cavay@fibertel.com.ar)
María B Lasala (mblasala@fibertel.com.ar)
Angela M Famiglietti (famiglie@ffyb.uba.ar)

Version: 6 Date: 7 April 2011

Author's response to reviews: see over
"Endocarditis caused by methicillin-susceptible Staphylococcus aureus with reduced susceptibility to vancomycin: a case report."

Beatriz E Perazzi, Natalia Bello, Marta Mollerach, Carlos A Vay, María B Lasala and Angela M Famiglietti
The changes are written in red in the revised manuscript.

Dear Dr. Deputy Editor:
According to your mail dated Monday, April 4, 2011:

The Deputy Editor would like you to make the following changes:

- The phrase “At admission, he presented febrile syndrome and chills”, should be “At admission, he presented with a febrile syndrome and chills”

Answer: The phrase was corrected in red as follows: "At admission, he presented with a febrile syndrome and chills".

We appreciate your valuable time in revising our work.

Yours faithfully
Professor Dr. Beatriz Perazzi